Title: Differences in the quality of primary medical care for CVD and Diabetes across the NHS: evidence from the Quality and Outcomes Framework

Authors:

Gary McLean (gml17y@clinmed.gla.ac.uk)
Bruce Guthrie (b.guthrie@chs.dundee.ac.uk)
Matt Sutton (m.sutton@abdn.ac.uk)

Version: 4 Date: 29 May 2007

Author's response to reviews:

Revised Manuscript: MS1117280466134288

Dear Editor,

We have enclosed our revised manuscript having performed the formatting changes requested.